Author's response to reviews

Title: The impact of chemotherapy-associated neutrophil/lymphocyte counts on the efficacy of adjuvant chemotherapy in colorectal cancer

Authors:

Chu-Yuan Hong (hongchuyuan@126.com)
Jing Peng (ykkhds1998@sina.com)
Yi-Sheng Wei (yswei2004@126.com)
He-Peng Peng (penghp@yeah.net)
Hui Yang (yanghui030454@gmail.com)
Chu-Xiong Zhao (zhaochux@126.com)
Guo-Jian Liang (854768420@qq.com)
Guo-Qiang Wang (wqjiyh@163.com)

Version: 2 Date: 2 April 2012

Author's response to reviews: see over
April 2, 2012

Christna Chap. Prof. Dr.
Executive Editor
BMC Cancer

Subject: The impact of chemotherapy-associated neutrophil/lymphocyte counts on the efficacy of adjuvant chemotherapy in colorectal cancer

Type of manuscripts: Original Article

Dear Professor Christna Chap.,

Thanks for inviting to revise our manuscript entitled “The impact of chemotherapy-associated neutrophil/lymphocyte counts on the efficacy of adjuvant chemotherapy in colorectal cancer” submitted to BMC Cancer for publication. We have made the changes on the manuscript as follows: We have included the title page and abstract in the main manuscript file, and the title page has contained Title, Author list, Affiliations (department names, institution name, street name, city, zip code, country) and email addresses.

Thank you very much for your consideration.

Yisheng Wei
PhD MD, Department of Gastrointestinal Surgery